# Peer support for discharge from inpatient to community mental health care: the ENRICH research programme

Steve Gillard,<sup>1\*</sup> Rhiannon Foster,<sup>2</sup> Sarah White,<sup>2</sup> Andrew Healey,<sup>3</sup> Stephen Bremner,<sup>4</sup> Sarah Gibson,<sup>2</sup> Lucy Goldsmith,<sup>2</sup> Mike Lucock,<sup>5</sup> Jacqueline Marks,<sup>2</sup> Rosaleen Morshead,<sup>2</sup> Akshaykumar Patel,<sup>6</sup> Shalini Patel,<sup>7</sup> Julie Repper,<sup>8</sup> Miles Rinaldi,<sup>7,9</sup> Alan Simpson,<sup>10</sup> Michael Ussher,<sup>2,11</sup> Jessica Worner<sup>12</sup> and Stefan Priebe<sup>13</sup>

#### **Disclosure of interests**

**Full disclosure of interests:** Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/LQKP9822.

Primary conflicts of interest: Stefan Priebe was listed a member of the HTA MPOH Panel from 1 January 2014 to 31 May 2018. Julie Repper's employer, ImROC, received payment for her work as a peer support expert consultant. Mr Rinaldi received travel and accommodation costs from Nordland Hospital/Research Council of Norway and is a member of the Data Monitoring Committee for the NIHR-funded ODDESSI trial. Alan Simpson is a co-director of the NIHR Mental Health Policy Research Unit. No other competing interests are declared.

<sup>&</sup>lt;sup>1</sup>School of Health Sciences, City, University of London, London, UK

<sup>&</sup>lt;sup>2</sup>Population Health Research Institute, St George's, University of London, London, UK

<sup>&</sup>lt;sup>3</sup>King's Health Economics, King's College London, London, UK

<sup>&</sup>lt;sup>4</sup>Department of Primary Care and Public Health, Brighton and Sussex Medical School, Brighton, UK

<sup>&</sup>lt;sup>5</sup>Centre for Applied Research in Health, University of Huddersfield, Huddersfield, UK

<sup>&</sup>lt;sup>6</sup>Pragmatic Clinical Trials Unit, Queen Mary, University of London, London, UK

<sup>&</sup>lt;sup>7</sup>Adult Community Mental Health Team, South West London & St George's Mental Health NHS Trust, London, UK

<sup>&</sup>lt;sup>8</sup>Implementing Recovery through Organisational Change, Nottingham, UK

<sup>&</sup>lt;sup>9</sup>Centre for Work and Mental Health, Nordland Hospital Trust, Bodø, Norway

<sup>&</sup>lt;sup>10</sup>Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK

<sup>&</sup>lt;sup>11</sup>Institute for Social Marketing and Health, University of Stirling, Stirling, UK

<sup>&</sup>lt;sup>12</sup>National Suicide Prevention Alliance, London, UK

<sup>&</sup>lt;sup>13</sup>Unit for Social and Community Psychiatry, Queen Mary, University of London, London, UK

<sup>\*</sup>Corresponding author steven.gillard@city.ac.uk

Published November 2023 DOI: 10.3310/LQKP9822

## Plain language summary

Peer support for discharge from inpatient to community mental health care: the ENRICH research programme

Programme Grants for Applied Research 2023; Vol. 11: No. 8 DOI: 10.3310/LQKP9822

NIHR Journals Library www.journalslibrary.nihr.ac.uk

## **Plain language summary**

When people share similar experiences of mental health problems and provide each other with emotional or practical support this can be called peer support. In recent years peer support has been introduced into mental health services.

The months after discharge from psychiatric hospital can be difficult for some people. There is research that suggests that peer support might be helpful for people at discharge. There is a lack of good quality trials of peer support in mental health services. There are no studies that tell us if peer support provides value for money.

We developed new training for peer workers and a handbook guiding peer support for discharge. We tried this out in two mental health National Health Service trusts to check that we could deliver the peer support and collect the information we needed for our trial.

We recruited 590 people from seven mental health National Health Service trusts while they were staying on psychiatric wards. Half of those people were randomly selected to receive peer support for discharge, and half to receive community mental health care only. We found that people offered peer support were just as likely to be readmitted to hospital in the year after they were discharged as people offered usual care only.

We found that people who met their peer worker at least twice were less likely to be readmitted in the next year. Black people offered peer support were much less likely to be readmitted compared to people offered usual care than other people in the study (although numbers were small). Our results also suggested that cost of care could be slightly less for people receiving peer support.

We interviewed some people who had been offered peer support who told us what they valued about peer support. We interviewed peer workers who told us the work was rewarding but could be challenging.

### **Programme Grants for Applied Research**

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

Programme Grants for Applied Research (PGfAR) was launched in 2013 and is indexed by Europe PMC, NCBI Bookshelf, DOAJ, Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and Scopus® (Elsevier, Amsterdam, Netherlands).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar.

#### Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **Programme Grants for Applied Research programme**

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health and Care Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

#### This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-1212-20019. The contractual start date was in March 2015. The final report began editorial review in May 2021 and was accepted for publication in September 2022. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the PGfAR programme or the Department of Health and Social Care.

Copyright © 2023 Gillard *et al.* This work was produced by Gillard *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

#### NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

#### NIHR Journals Library Editors

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

**Dr Peter Davidson** Interim Chair of HTA and EME Editorial Board, Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Dr Rob Riemsma** Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

**Professor Helen Roberts** Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk